Author’s response to reviews

Title: Validation study of the Chinese Early Development Instrument (CEDI)

Authors:

Patrick Ip (patricip@hku.hk)
Ling Li (sophia.ling@gmail.com)
Nirmala Rao (nrao@hku.hk)
Sharon Sui Ngan Ng (suinng@ied.edu.hk)
Winnie Wai Sim Lau (winniewslau@hotmail.com)
Chun Bong Chow (chowcb@netvigator.com)

Version: 2 Date: 7 September 2013

Author’s response to reviews: see over
Dear Professor O’Donovan,

Many thanks for forwarding the reviewer’ comments concerning our paper. We found them to have been extremely helpful and encouraging. Below is an account of what we have done to improve the paper based on the reviewer’ suggestions. Please note we have not ignored any points. We have also made the mandatory style changes to meet the style of the journal: BMC Pediatrics.

We thank for your assistance in improving our manuscript. Looking forward to your favorable reply soon.

Sincerely,
Dr Patrick Ip